Author’s response to reviews

Title: Comparative efficacy of serotonin (5-HT3) receptor antagonists in patients undergoing surgery: A systematic review and network meta-analysis

Authors:

Andrea C Tricco (TriccoA@smh.ca)
Charlene Soobiah (SoobiahC@smh.ca)
Erik Blondal (BlondalE@smh.ca)
Areti A Veroniki (VeronikiA@smh.ca)
Paul A Khan (pakahn@hotmail.com)
Afshin Vafaei (VafaeiA@smh.ca)
John Ivory (john_d_ivory@gmail.com)
Lisa Strifler (StriflerL@smh.ca)
Huda Ashoor (AshoorH@smh.ca)
Heather MacDonald (hrmacdonald@gmail.com)
Emily Reynen (ereynen@gmail.com)
Reid Robson (reidcrobson@gmail.com)
Joanne Ho (jmho2001@gmail.com)
Carmen Ng (carmen_hm_ng@yahoo.ca)
Jesmin Antony (AntonyJ@smh.ca)
Kelly Mrklas (kmrklas@gmail.com)
Brian Hutton (bhutton@ohri.ca)
Brenda R Hemmelgarn (Brenda.Hemmelgarn@albertahealthservices.ca)
David Moher (dmoher@ohri.ca)
Sharon E Straus (sharon.straus@utoronto.ca)

Version: 2
Date: 11 May 2015

Author’s response to reviews:

Dear BMC Med,

Since the first author is an Assistant Editor for the BMC Med Research Method, and the last author is an Assistant Editor for Implementation Science (also a BMC journal), we were wondering if we could have a discount on our paper?

Thank you for considering,
Andrea Tricco